Author's response to reviews

Title: Increased expression of urokinase plasminogen activator and its cognate receptor in human seminomas

Authors:

Salvatore Ulisse (salvatore.ulisse@uniroma1.it)
Enke Baldini (baldinienke@virgilio.it)
Marcella Mottolese (mottolese@ifo.it)
Steno Sentinelli (steno8@virgilio.it)
Patrizia Gargiulo (patrizia.gargiulo@uniroma1.it)
Salvatore Sorrenti (salvatore.sorrenti@uniroma1.it)
Anna Di Benedetto (dibenedetto@ifo.it)
Massimino D'Armiento (massimino.darmiento@uniroma1.it)
Enrico De Antoni (enrico.deantoni@uniroma1.it)
Brancato Valentina (valentina.brancato@gmail.com)

Version: 3 Date: 31 March 2010

Author's response to reviews:

Re: Ms # 2025934211324547 (Re-Revised)

Title: "Increased expression of urokinase plasminogen activator and its cognate receptor in human seminomas"

Dear Editor,

Thank you for handling the manuscript and for your letter. I'm glad to learn that the manuscript is now acceptable for publication and I'm grateful to the Reviewers for their comment and work on the manuscript. The only revision required by the Reviewer 3 has been addressed. In particular, with add the year of publication of the reference 17 as indicated by the Reviewer.

Thank you again for your work on the manuscript and I look forward to hear from you soon.

Yours sincerely,

Prof. Massimino D'Armiento
Dept. of Experimental Medicine
"Sapienza" University of Rome
Rome-Italy
Tel.: (+39)-06-49972601
FAX: (+39)-06-49972606
e-mail: massimino.darmientoniroma1.it